Author's response to reviews

Title: Prophylactic effect of HESA-A on influenza A virus infectivity

Authors:

Parvaneh Mehrbod (mehrbode@yahoo.com)
Aini Ideris (mehrbode@gmail.com)

Version: 2 Date: 21 September 2013

Author's response to reviews: see over
21 September, 2013

Reply to the Remarks of the Editor

Dear Editor,

BMC Complementary and Alternative Medicine (Section: Basic research)

Thank you for your email on 20 September, 2013 regarding manuscript MS: 9532243410858951. We have revised the previous version of MS very carefully and made the appropriate corrections. The changes to the text and reply to the editor’s remarks by an itemized list of responses is enclosed.

1. The sentence regarding vaccine overlooking was removed from abstract.
2. The concluding sentence regarding HESA-A recommendation was moderated.
3. As I noticed, everywhere I have mentioned 2 authors names: (Parvaneh Mehrbod and Aini Ideris): after title, authors contributions and online submission.

Thank you,

Corresponding Author
Dr. Parvaneh Mehrbod
Laboratory of vaccine and immunotherapeutic
Institute of Bioscience
Universiti Putra Malaysia
43400 UPM Serdang,
Selangor, Malaysia.
Email: mehrbode@yahoo.com